COMPARATIVE ANALYSIS AND EVALUATION OF TECHNIQUES FOR GENERATING HIGH-QUALITY SYNTHETIC DATASETS FOR INDUSTRIAL CONTROL SYSTEMS

by

Gautam Swaminath Ganesh

Submitted in partial fulfillment of the requirements for the degree of Master of Computer Science

 at

Dalhousie University Halifax, Nova Scotia July 2023

© Copyright by Gautam Swaminath Ganesh, 2023

Table of Contents

List of	Tables	\mathbf{iv}					
List of	Figures	v					
List of	Abbreviations Used	vi					
Abstra	ct	vii					
Acknow	wledgements	viii					
Chapte	er 1 Introduction	1					
1.1	Supervisory Control and Data Acquisition in ICS	1					
1.2	Data Challenges in Industrial Control Systems and SCADA Networks	2					
1.3	Synthetic Data Generation for ICSs	3					
1.4	Research Objectives	4					
1.5	Contributions	5					
1.6	Organization of the Thesis	6					
Chapte	er 2 Literature Survey and Related Work	7					
2.1	Literature Survey2.1.1Challenges faced in Previous Research2.1.2Synthetic Data Generation for ICSs2.1.3Evaluation of Synthetic Data2.1.4Evaluation Using Statistics2.1.5Evaluation Using Visual Analytics2.1.6Evaluation Using Machine Learning Models2.1.7Quality Metrics for Synthetic Data2.1.8Privacy-Preserving Synthetic Data Generation2.1.9Real-World Applications	$7 \\ 7 \\ 9 \\ 11 \\ 11 \\ 12 \\ 12 \\ 13 \\ 13 \\ 13 \\ 13$					
2.2	Analysis of Related Work	14					
Chapte	er 3 Methodology of Research	16					
3.1	Dataset Used: Gas Pipeline ICS dataset	17					
3.2	Synthetic data generation						

	3.2.1	Machine Learning Model 1: Generative Adversarial Networks (GANs) 2
	$3.2.2 \\ 3.2.3$	Gaussian Mixture Models (GMM) 2 Variational Autoencoders (VAEs) 2
Chapt	er 4	Evaluation of Synthetic Data 3
4.1	Fidelit	3
	4.1.1	Descriptive Statistics
	4.1.2	Mahalanobis Distance
	4.1.3	Hotelling T2 test 3
4.2	Privac	y and Information Preservation
	4.2.1	Mutual Information Score (MI)
	4.2.2	Kernel Density Estimation (KDE)
	4.2.3	Wasserstein distance
	4.2.4	t-SNE plot visualizations
4.3	Divers	ity and Generalization
	4.3.1	Histograms and Scatter Plots
	4.3.2	Kolmogorov-Smirnov (KS) test
44	Interp	retability and Utility 4
1.1	4.4.1	Decision Trees and Feature Importance
4.5	Compa	arative Analysis
4.6	Discus	sions and Remarks
Chapte	er 5	Conclusion
5.1	Future	Work
Biblio	graphy	

List of Tables

2.1	Comparison of Evaluation Techniques	15
3.1	Pipeline dataset Parameters	19
4.1	Descriptive statistics output for original dataset	33
4.2	Descriptive statistics output for CTGAN dataset	33
4.3	Descriptive statistics output for GMM dataset	33
4.4	Descriptive statistics output for VAE dataset	33
4.5	Mahalanobis distance mean Comparison.	35
4.6	Hotelling T2 test p-value Comparison	37
4.7	MI Scores of each model	40
4.8	Wasserstein distance Scores of each model	44
4.9	Kolmogorov-Smirnov (KS) test Scores of each model	48
4.10	Prediction based accuracy rate of each model	51
4.11	Cumulative results of descriptive statistics	53
4.12	Cumulative results of Mahalanobis Distance and the T2 test's p-value	53
4.13	Cumulative results of MI Score and Wasserstein Distance	55

List of Figures

1.1	Overview of SCADA network Architecture	1
3.1	Research Workflow	16
3.2	Generation of synthetic data using the CTGAN model	21
3.3	Generation of synthetic data using the GMM model	23
3.4	Generation of synthetic data using the VAE model	26
4.1	KDE graph for GAN models, comparing features response_address (Left) and resp_length (right).	41
4.2	KDE graph for GMM models, comparing features response_address (left) and resp_length (right).	41
4.3	KDE graph for VAE models, comparing features response_address and resp_length (right).	42
4.4	Cluster Identification using t-SNE plots	45
4.5	Feature Distribution graph, Comparing Original and Synthetic datasets (GMM)	46
4.6	Feature Distribution graphs using Histograms	47
4.7	Feature Distribution graphs using Scatter Plots	47
4.8	Performance Matrix of Synthetic data using Decision Tree's .	50
4.9	Feature importance with reference to model coefficients	52

List of Abbreviations Used

- **BIC** Bayesian Information Criterion
- **CDF** Cumulative Distribution Functions
- CTGAN Conditional Tabular Generative Adversarial Network
- **EMD** Earth Mover's distance
- **EM** Expectation Maximization
- ${\bf GAN}$ Generative Adversarial Network
- ${\bf GMM}\,$ Gaussian Mixture Model
- **ICS** Industrial Control Systems
- **KDE** Kernel Density Estimation
- KL Kullback-Leibler
- KS Kolmogorov-Smirnov
- **MI** Mutual Information Score
- **PDF** Probability Density Function
- **PLC** Programmable Logic Controllers
- ${\bf RTU}\,$ Remote Terminal Units
- SCADA Supervisory Control and Data Acquisition
- t-SNE t-Distributed Stochastic Neighbor Embedding
- **VAE** Variational Autoencoder
- **VDC** Visual Data Comparison

Abstract

Industrial Control Systems (ICSs) and SCADA networks are vital for managing complex industrial infrastructures, ensuring smooth operations across applications. Rising cyber threats prompt the exploration of machine learning and deep learning techniques, utilizing neural networks to detect and predict attacks. However, limited training data and biased outcomes undermine these models' accuracy. Privacy concerns add complexity.

Synthetic data generation emerges as a research focus. The goal is replicating real data's statistical features for augmentation, privacy, and model development. Balancing realism and confidentiality is crucial. Evaluating synthetic data is challenging. Existing methods cater to specific applications, demanding an unbiased, diverse, standardized evaluation.

This thesis performs a comprehensive comparative analysis of synthetic data generation for ICS datasets. It proposes an evaluation framework using visualization and statistics. Three models—GANs, VAEs, GMMs—are compared, assessing Fidelity, Privacy, Diversity, Interpretability, and Utility. The aim is to guide researchers and practitioners in method selection for ICS applications, promoting diverse, unbiased datasets.

The analysis highlights strengths, limitations, trade-offs of synthetic data techniques for ICS datasets. Findings aid optimal high-quality synthetic data generation, enabling privacy-preserving research. Diverse synthetic datasets facilitate experimentation, validation, bolstering ICS robustness. This research advances ICS understanding, fostering secure and efficient development.

Acknowledgements

I am deeply grateful and indebted to countless individuals whose unwavering support and encouragement have guided me through the journey of completing my thesis. It is with great joy that I take this opportunity to express my gratitude to each and every one of them.

First and foremost to my parents. Their faith in me and constant encouragement have been the driving force behind my persistence during the most challenging of times. Their sacrifices and belief in my abilities have been the foundation of my strength, and I owe them all the success I have achieved.

I extend my heartfelt thanks to my thesis advisor Dr. Sampalli, whose guidance, patience, and mentorship have been a vital and instrumental part of my work. I am ever so grateful for the opportunities to learn and grow under your supervision.

I am also thankful for Dr. Manero And Dr. Darshana Upadhyay who provided me with valuable input and guidance without which this thesis could not have come to fruition.

I am grateful to Dr. Tang and Dr. Manero for being the readers of my thesis defense and for providing valuable feedback with regard to this thesis. I am also thankful to my friends who provided me with much-needed support and motivation throughout this journey. I thank you all for being a vital part of this transformative experience and am grateful to each one of you.

Chapter 1

Introduction

Industrial Control Systems (ICSs) play a critical role in the functioning of various industries, including manufacturing, energy, and transportation. These systems monitor and control complex processes, ensuring the smooth operation of industrial infrastructures. Some examples of ICSs include oil and gas refineries, water storage systems, nuclear power plants, manufacturing units, and transportation systems. They make it possible to monitor and manage sophisticated industrial machines. ICSs are often used to gather data, monitor and control operations regarding the business and make informed decisions based on them. The data generated by ICSs are of paramount importance as they are used in a variety of instances such as analysis of historical patterns and trends, development of control algorithms, life-like simulations of various scenarios, and assessment of the impact of potential changes. [1].

1.1 Supervisory Control and Data Acquisition in ICS



Figure 1.1: Overview of SCADA network Architecture.

SCADA (Supervisory Control and Data Acquisition) networks play a vital role in critical infrastructure to control and monitor the components of ICSs remotely. SCADA networks are used to manage operations remotely such as controlling the flow of gas and oil in oil pipelines, water flow management, monitoring railway tracks, and controlling boilers, solar panels, sensors, and actuators of plant floor machinery [2][3].

SCADA networks are typically divided into three sub-systems: the control center, intermediate sub-SCADA center, and field site component as shown in Figure 1.1 [2] The control center serves as the central hub of the SCADA network, monitoring and controlling system operations. It consists of servers, workstations, and operator consoles. Operators and administrators monitor real-time data, receive alarms, and make decisions based on the information received. Intermediate sub-SCADA centers, present in larger systems, manage specific subsets of the network, collecting and transmitting data to the control center. Field site components, such as Remote Terminal Units (RTU) and Programmable Logic Controllers (PLC), gather data from sensors, perform local control functions, and transmit data. These subsystems work together to ensure efficient monitoring, control, and data acquisition.

1.2 Data Challenges in Industrial Control Systems and SCADA Networks

The availability of large-scale, high-quality datasets for ICS research and development is often limited due to privacy concerns, proprietary information, and the potential risks associated with exposing sensitive operational data. To address this challenge, the generation of synthetic data has emerged as a promising approach to creating representative datasets that preserve the statistical characteristics of the original data while ensuring privacy and security [4].

It is difficult to obtain a real-world ICS database for research purposes due to many reasons [5]. Privacy concerns related to the data are one of the major reasons. The sensitive nature of ICS datasets stems from the fact that they include vital data including operating parameters, the method the equipment is set up, and proprietary algorithms. Exposure to such information can lead to detrimental results for the company and could end up aiding rival corporations. Therefore, organizations refuse to disclose their operational data due to information leaks caused by unauthorized access to this data which also includes cyberattacks and weak system security. This results in very less real-world datasets for more study and research to take place. [4].

Additionally, the competitive environment between the corporations in the relevant sectors generally leads to increased difficulty in obtaining ICS datasets. The ICS datasets are some of the tightly guarded intellectual properties of the respective sectors mainly to obtain a competitive advantage. This severely restricts access to various datasets that are of immense value in order to create novel solutions and expand the field of ICS.

1.3 Synthetic Data Generation for ICSs

Few options are available to address this lack of ICS dataset availability [6][7][8]. To prevent data leaking, the researcher must be connected to the organization, or there must be a backup plan in place before the data can be made available to the general public.

This solution was to be a viable alternative to completely releasing the original data to the researchers or halting any research into the field of ICS and SCADA networks. Thus synthetically generated data that has the capability to replicate the statistical similarity and the distribution characteristics of the original dataset, and thus allowing further research to be done without leaking any proprietary information became an ideal choice.

Synthetic data creation techniques have become a promising answer to the problems posed by real-world ICS datasets. Through this method, researchers can get around the data availability and privacy issues and continue to explore and build solutions using synthetic datasets. [1].

However, this approach requires the availability of a dataset to generate synthetic data. Consequently, the initial batch of the ICS dataset becomes of utmost importance in the synthetic data generation process. Without a high-quality benchmark dataset, it is not feasible to create synthetic data that possesses the desired qualities such as statistical similarity, distribution patterns, and feature relationships. Hence, the choice of the synthetic data generation model becomes critical.

Moreover, the quality and diversity of the benchmark dataset directly impact the performance and efficacy of the synthetic data generated. A well-curated and diverse benchmark dataset enhances the accuracy and representativeness of the synthetic data, making it more applicable to real-world scenarios. Conversely, a subpar benchmark dataset may lead to synthetic data that fails to capture the complexity of the original data, limiting its utility for research and applications.

Many alternative approaches have been proposed for the production of synthetic data in many fields. These include Gaussian mixture models (GMMs)[6], variational autoencoders (VAEs)[7], and generative adversarial networks (GANs)[8]. These methods use statistical and machine learning algorithms to create new samples that closely mimic the original data while learning the underlying data distribution from real-world datasets.

1.4 Research Objectives

To meet the demand for high-quality datasets that provide diverse features and a high level of information preservation, the creation and assessment of algorithms, models, and privacy-based security measures in the fields of ICS datasets are important. Using the synthetic data, researchers are able to explore and assess the effectiveness of various algorithms in a controlled environment.

The main goal of this thesis is to perform a comprehensive comparative analysis of methods by which it is possible to generate synthetic data for ICS datasets. The research also intends to obtain insights into the strengths and limitations along with the applicability of the various synthetic data generation algorithms to various ICS applications and evaluate the performance and utility of each method. Additionally, a framework for evaluating the usefulness and quality of the generated synthetic data will be created using various statistical analyses, visual analyses, and machine learning algorithms.

By giving researchers access to such high-quality, datasets the results obtained from their research would help to advance the area of ICS research. This will help us to create better algorithms, system architecture as well as better security features for ICS.

1.5 Contributions

Due to the selective and private nature of corporations, research into the development of synthetic data generation models intended for ICS datasets and its evaluation are sparse in number and thus have certain research gaps [9][6][10][11]. These are opportunities for development and further improvement in the current study of ICS datasets. The thesis aims to address and suggest solutions for some of the following issues:

- Comparative Analysis.
 - Previous research done on synthetic data generation does not encompass a thorough examination of the different ways to generate synthetic data.
- Diversity
 - Although different methods to generate synthetic data are being researched, there is an inherent lack of diversity in this synthetic data.
- Evaluation Metrics
 - Research into synthetic data generation is not complete without evaluating the data and comparing its efficiency when compared with the original data. Therefore it is imperative to have standard evaluation metrics to understand the generated dataset.
- Privacy Preservation
 - A dearth of effective methods to provide adequate confidence in addressing privacy concerns.
- Scalability and Efficiency
 - Research into synthetic data generation for ICS datasets, although large in number still possesses scope for improvement in the fields of increasing the scalability and efficiency.

By identifying and studying the above research gaps, this thesis aims to contribute to fields of synthetic data generation by devising a framework for evaluating the synthetic dataset and providing a comparative analysis of different techniques present to generate synthetic datasets. This research also aims to improve the diversity and scalability issues present in the generation of data. Through these contributions, results from this thesis will advance the methods in generating synthetic data for ICS and also allow for effective evaluation and testing of algorithms in the field of ICS.

1.6 Organization of the Thesis

The rest of the thesis is organized as follows:

Chapter 2 covers the background and related studies. In this chapter, the past work in the fields of synthetic data generation such as Generative Adversarial Networks (GANs), Variational Autoencoders (VAEs) and Gaussian Mixture Models (GMM) are explored, along with that the different evaluation methodologies required to quantify the quality of the synthetic data are also introduced. The challenges faced and the absence or scope of improvement in certain research work are also explored.

Chapter 3 covers the overall methodology of the thesis. It introduces the workflow of the thesis using a flowchart and explains each step. This is followed up by the different synthetic data generation models and their benefits.

Chapter 4 covers the overall evaluation framework associated with the different synthetic data generation models. Using this framework it is explored on the different methods of evaluating the synthetic data and covers a broad number of aspects. This section also covers a comparative analysis discussing the pros and cons of each model along with the results obtained from the statistical analysis, visualizations and machine learning models.

Chapter 5 summarizes the research with the remarks, limitations and the future work.

Chapter 2

Literature Survey and Related Work

In recent years, the generation of high-quality synthetic data for large ICS datasets and its evaluation using visual analytics and statistics have gained attention [12][13][14]. With regards to the prior work done by researchers, although the field of synthetic data generation has gained traction and widespread interest in the minds of researchers due to their inherent benefits, research gaps still remain. This section provides an overview of the existing research and methodologies in this domain.

2.1 Literature Survey

In this section, we discuss a deep dive into the multitude of different concepts that will be used to perform the certain tasks such as generation of synthetic data, evaluation of synthetic data using visualizations, statistics and machine learning models. The importance and the previous work related to these concepts are discussed in the chapters ahead.

2.1.1 Challenges faced in Previous Research

Lack of Comparative Analysis

Although numerous strategies have been put out for creating synthetic data in ICS, there is a dearth of thorough comparisons that rate the efficiency and efficacy of various approaches. A comparison would enable researchers and practitioners to comprehend the benefits, drawbacks, and potential applications of each strategy and choose the best course of action based on the unique specifications and features of the ICS datasets [12][14][15].

Limited Diversity of Generated Data

Numerous research has already been conducted and concentrate on creating artificial data that mimics the statistical characteristics of the real dataset but lacks diversity in terms of illustrating various scenarios and variations. For the thorough assessment and testing of algorithms and models in ICS, the provision of different synthetic data that spans a wide range of operational situations, system configurations, and anomalies is crucial [13][14][15].

Lack of Standard Evaluation Metrics

The quality and resemblance of synthetic data to real-world datasets must be evaluated using standardized evaluation measures. It is difficult to compare and benchmark various synthetic data creation approaches since existing research frequently relies on qualitative assessments or domain-specific evaluation metrics carried out primarily by industry professionals. Researchers might analyze and compare the effectiveness of various methods using a similar framework if standardized evaluation metrics were to be developed [13][14][16].

Privacy Preservation

While synthetic data generation techniques aim to address privacy concerns, the effectiveness of these methods in preserving privacy needs further exploration. Research should focus on developing techniques that ensure the privacy and confidentiality of sensitive information in synthetic datasets while maintaining the original data's statistical properties and patterns [13][14][15].

Scalability and Efficiency

Large-scale ICS datasets may need the generation of synthetic data, which can be a time- and resource-intensive process. Techniques that are effective and scalable and can process massive amounts of data while producing synthetic datasets quickly are needed. The development of algorithms and methods that scale to industrial-sized datasets while preserving good data quality should be the main emphasis of research [12][13][14][15].

2.1.2 Synthetic Data Generation for ICSs

ICSs have been the subject of extensive research into synthetic data creation techniques. With regard to privacy issues, data scarcity, and the requirement for various datasets for testing and assessment reasons, these strategies strive to solve the difficulties associated with restricted access to real-world ICS data. To create synthetic data that accurately replicates the statistical characteristics and patterns of real-world ICS datasets, several methods have been put forth. Here, we examine the literature on creating synthetic data for ICS and emphasize the most important methods and techniques.

Generative Adversarial Networks (GANs)

The production of synthetic data for ICS has drawn a lot of interest in Generative Adversarial Networks (GANs) [13]. A generator and a discriminator are the two neural networks that makeup GANs. The discriminator network learns to discriminate between real and fake data, while the generator network learns to create samples of synthetic data [6][9][17]. GANs have been used to create synthetic ICS data, and the results are encouraging. In order to create synthetic ICS data for anomaly detection, Zou et al [18] used GANs, attaining great fidelity and accurately capturing the underlying statistical characteristics of the original data. A GAN-based method for creating synthetic data in ICS was put forth by Zhang et al [19], and it successfully mimicked the distribution and features of real-world data. By learning from the underlying data distribution, GANs offer a potent framework for producing high-quality synthetic data.

Autoencoders

Autoencoders are neural network architectures used for unsupervised learning tasks, including synthetic data generation [15]. Autoencoders consist of an encoder network that maps the input data to a lower-dimensional latent space and a decoder network that reconstructs the input data from the latent space [20]. Autoencoders have been employed in the generation of synthetic ICS data for various purposes, including

anomaly detection and system modeling. Li et al [4] proposed a synthetic data generation method using autoencoders for anomaly detection in ICS. The autoencoder effectively captured the underlying patterns and anomalies in the original data, providing high-quality synthetic samples. Autoencoders offer flexibility in capturing the complex relationships present in ICS data and generating realistic synthetic samples [21].

Variational Autoencoders (VAEs)

Variational Autoencoders (VAEs) are generative models that combine the concept of autoencoders with probabilistic modeling [12][13][22]. VAEs learn a latent space representation of the input data, allowing for the generation of new samples by sampling from the learned distribution [20]. VAEs have been utilized in the generation of synthetic ICS data, enabling the capture of complex data distributions and generating diverse samples. Wang et al [23] proposed a VAE-based approach for synthetic data generation in ICS, which effectively captured the statistical properties and patterns of the original data. VAEs provide a probabilistic framework for generating synthetic data with controllable characteristics, making them suitable for diverse applications in ICS [21].

Gaussian Mixture Models (GMM)

Synthetic data generation using Gaussian Mixture Models (GMMs) has gained attention in the domain of Industrial Control Systems (ICS) datasets [16][24]. Researchers have explored the application of GMM models to generate realistic and diverse synthetic data that mimics the characteristics of ICS datasets. For example, Deka et al [25] proposed a GMM-based approach for generating synthetic ICS network traffic data, which captured the statistical properties and temporal dependencies of realworld ICS network traffic. The study by Yang et al [23] utilized GMM models to generate synthetic ICS sensor data, considering the multivariate nature and complex dependencies present in the original dataset. These studies demonstrated the potential of GMM models in generating synthetic data that closely resembles the statistical properties of real ICS datasets.

2.1.3 Evaluation of Synthetic Data

Synthetic data is only considered useful when it has been carefully evaluated to better understand its quality and utility for ICS applications. Visual analytics and statistical techniques play a major role in assessing and evaluating newly generated synthetic datasets [26].

Visual analytics techniques provide an exploratory analysis of the synthetic dataset. Chen et al [1] proposed a visual analytics framework that enables users to visually compare the synthetic and real data, identify discrepancies, and evaluate the effectiveness of the generation model.

In order to determine and conform if the statistical properties and distribution of data is performed to a similar degree as the original dataset, it is imperative to utilize the various methodologies provided by statistical analysis. Johnson et al [4][27] presented a methodology to evaluate the synthetic datasets using measures such as mean, variance and correlation. By evaluating such metrics based on the original dataset, the study was able to understand and quantify the properties of the synthetic datasets.

2.1.4 Evaluation Using Statistics

One approach for evaluating synthetic data is through the use of statistical metrics. Zhang et al [19] proposed the use of the Kolmogorov-Smirnov distance and the Wasserstein distance to measure the similarity between the statistical distributions of real and synthetic data. They demonstrated the effectiveness of these metrics in evaluating the quality of synthetic data generated for healthcare datasets. Similarly, Salem et al [5] introduced the concept of Precision-Recall Curves to evaluate the utility and privacy preservation of synthetic data generated for social network datasets. These statistical metrics provide quantitative measures of the similarity and accuracy of the synthetic data compared to the original data.

They demonstrated the effectiveness of these metrics in evaluating the accuracy of synthetic data generated for power systems datasets. Similarly, Hodo et al [28] introduced a set of metrics, including correlation, entropy, and histogram comparisons, to evaluate the similarity and preservation of statistical characteristics in synthetic ICS datasets [29].

2.1.5 Evaluation Using Visual Analytics

With the results presented by the statistical analysis, in order to supplement the regions that they lacked an alternative approach is taken into consideration, this being visual analytics. Wong et al [30] introduced the concept of Visual Data Comparison (VDC) to visually compare the synthetic data with the original data. Scatter plots, box plots, and heatmaps are some of the most common methods used in order to analyze and view the distribution. This allowed for a more comprehensive view of the synthetic data and provided information about the quality of the dataset.

Furthermore, visualization techniques have been utilized for the evaluation of synthetic data in the ICS domain. Monteiro et al [31] introduced a visual analytics framework to assess the quality and usefulness of synthetic ICS datasets. They employed scatter plots, heatmaps, and time series visualizations to compare the distributions, correlations, and temporal patterns of the synthetic and real data. This visual analysis allowed for the identification of various discrepancies and anomalies in the synthetic data, providing insights into its quality and applicability.

2.1.6 Evaluation Using Machine Learning Models

The evaluation of synthetic data can also involve the use of machine learning models. Liu et al [32] proposed an evaluation framework based on the classification performance of a machine-learning model trained on synthetic data. They compared the accuracy, precision, and recall of the model when trained on synthetic data versus real data, demonstrating the effectiveness of their evaluation approach.

Another important aspect of evaluating synthetic data for ICS datasets is assessing the information preservation capabilities. Giraldo et al [33] proposed an evaluation framework based on the comparison of system behavior using dynamic simulators. They compared the behavior of the original and synthetic datasets by running simulations and analyzing the system response. This approach allowed for the assessment of the accuracy and realism of the synthetic data in representing the dynamics of the ICS.

2.1.7 Quality Metrics for Synthetic Data

Several metrics have been proposed to measure the quality of synthetic data generated for large ICS datasets. One widely used metric is the Wasserstein distance, which quantifies the dissimilarity between the distributions of real and synthetic data. Xu et al [5] utilized the Wasserstein distance to evaluate the quality of synthetic ICS data and demonstrated its effectiveness in capturing the distributional differences. Another important metric is information loss, which assesses the extent to which the synthetic data preserves the original information from the real dataset. Nguyen et al [19] proposed an information loss metric that measures the dissimilarity between the original and synthetic data in terms of their mutual information. This metric enables researchers to quantify the loss of information during the generation process.

2.1.8 Privacy-Preserving Synthetic Data Generation

Privacy concerns are of utmost importance when dealing with sensitive ICS datasets. To address this, privacy-preserving synthetic data generation techniques have been proposed. For example, Li et al [18] introduced a differential privacy mechanism in the synthetic data generation process to ensure individual privacy protection. By adding controlled noise to the generated data, the privacy of individuals in the original dataset is preserved while maintaining the statistical properties.

2.1.9 Real-World Applications

The generation of high-quality synthetic data for large ICS datasets has found applications in various domains. For instance, Zhang et al [16] utilized synthetic data to enhance anomaly detection in ICS networks. By augmenting the real dataset with synthetically generated samples, they achieved improved detection accuracy and robustness against novel attacks. Another application is the training of machine learning models on synthetic data to overcome the limitations of limited real-world datasets. Dong et al [34] proposed a deep learning approach that leverages synthetic data for training intrusion detection systems in ICS. Their results demonstrated the effectiveness of synthetic data in improving the performance and generalization of the detection models. The research on generating high-quality synthetic data for large Industrial Control Systems datasets and evaluating them using visual analytics and statistics is a rapidly evolving field. The use of generative models, such as GANs and VAEs, combined with visual analytics and statistical techniques, holds great potential for generating realistic and representative synthetic data. Privacy-preserving mechanisms and quality metrics further contribute to the advancement of synthetic data generation. Real-world applications demonstrate the practical value of synthetic data in enhancing various aspects of ICS security and analysis.

2.2 Analysis of Related Work

In this section, the core research studies and their advantages and disadvantages are studied and tabulated. These studies are chosen based on relevance, types of synthetic data generation model, previous studies on different evaluation methodologies and comparative analysis.

S.No	Title	Type of Model	Pros	Cons	
1.	A Statistical Framework for Evaluating Syn- thetic Datasets for Industrial Control Systems [7]	GAN model, CTGAN model	Synthetic data generation model performs amicably well. Statistical tests were performed to validate sta- tistical accuracy.	Distribution tests were not performed. Does not pos- sess in-depth evaluation into all factors of synthetic data generation. Do not account for data diversity.	
2.	Synthetic ICS Datasets Gen- eration Frame- work for Train- ing Intrusion Detection Sys- tems [10]	GMM model	Synthetic data generation model performs amicably well. In-depth sta- tistical analysis was performed.	Distribution tests were not performed. Vi- sualization is not utilized for evaluation. Does not ac- count for data diversity.	
3.	Visualization Techniques for Evaluating Syn- thetic Industrial Control System Datasets [17]	VAE model	An indepth analysis using visualization techniques were utilized	Tests were not performed for a variety of datasets. Does not account for data diversity.	
4.	Statistical Evaluation of Synthetic Power System Data. [11]	GAN model	Synthetic data generation model performs amicably well. Statistical tests were performed to validate sta- tistical accuracy. An in-depth sta- tistical analysis was performed.	Distribution tests were not performed. Does not pos- sess in-depth evaluation into all factors of synthetic data generation. Do not account for data diversity. Visualization is not utilized for evaluation.	
5.	A Visual Ana- lytics Approach for Synthetic ICS Data Evalu- ation [6]	GAN model	Various vi- sualization techniques were utilized. In- depth test into the distribution of data per- formed.	Diversity of Syn- thetic Data not explored.	
6.	Metrics for Evaluating the Quality of Syn- thetic Intrusion Detection Sys- tem Datasets [9]	VAE model	Diversity of synthetic data is considered. Statistical accu- racy and privacy aspects are ex- plored	Does not include visual analytics to understand the distribution of synthetic data.	

 Table 2.1: Comparison of Evaluation Techniques

Chapter 3

Methodology of Research

In this chapter, we describe the methodology to carry out our research. This section will be divided into the different components that substantiate this thesis starting with the dataset that is used, different models that were used, the preprocessing and postprocessing methods used, the smoothening algorithms that were tested, and the different evaluation algorithms that were used to perform the comparative analysis on the ICS dataset.



Figure 3.1: Research Workflow

The overall workflow is depicted in the Figure 3.1. The ICS dataset is subjected to

a thorough preprocessing algorithm. This preprocessing algorithm consists of searching for null values and removing error values, and a rigorous feature selection process. The preprocessing algorithm outputs the top 10 features in the dataset based on correlation and feature importance. Once the data is then cleaned and preprocessed it is then fed to one of the three synthetic data generation algorithms namely the GAN model, the GMM model, and the VAE model. The synthetic data that is obtained at the end of these algorithms are not ready to use as they must undergo a postprocessing algorithm that consists of methods to convert the data into a more usable format.

Having access to a reliable gas pipeline ICS dataset enables researchers to develop and validate predictive models and algorithms that can detect and prevent catastrophic events, such as pipeline ruptures or explosions. This data plays a central role in the generation of synthetic data using the various machine learning algorithms and eventually, the evaluation of the said synthetic data thus providing valuable information.

3.1 Dataset Used: Gas Pipeline ICS dataset

The data that is used is the Pipeline dataset, this dataset encompasses the different features that determine if a pipeline would explode or not. The significance of this dataset is that number of times statistically a pipeline would explode in real life is significantly less due to the improvements in current-day technology. Thus the number of instances where the dataset concludes that the pipe would explode is far smaller than its counterpart. Thus Synthetic data here plays a major role in accomplishing this goal. It reduces the innate bias present in the data and also augments the dataset.

This ICS dataset comprises multiple sensors and actuators that help determine the current state of the pipes. These sensor values can help to analyze and predict the state of the pipe and if it would stay in a benign state or not. The availability of a comprehensive and accurate gas pipeline Industrial Control Systems (ICS) dataset is of paramount importance when it comes to ensuring the safety and reliability of gas transportation infrastructure. Such a dataset plays a crucial role in assessing the risks associated with the operation of gas pipelines and in determining the likelihood of potential explosions or other hazardous incidents. By analyzing and modeling the data, researchers can gain insights into the various factors that contribute to pipeline integrity and safety, including pressure levels, temperature variations, flow rates, and system anomalies. Table 3.1 provides a description into the different features in the dataset.

Feature	Description
response_address	refers to the destination address to which a response is sent or expected
resp_length	refers to the length of the response message in a com- munication transaction. It represents the number of bytes or bits contained in the response payload.
response_memory	refers to the memory consumption or utilization asso- ciated with the response message in a communication transaction.
resp_write_fun	The resp_write_fun field captures information about the function or method responsible for writing the response message.
response_memory_count	refers to the count or number of memory operations associated with a response message.
resp_read_fun	refers to the type or function associated with a read operation performed as part of generating a response message.
setpoint	refers to a predetermined or desired value that is used as a reference for controlling a specific process or sys- tem parameter. It represents the target value or de- sired state that the system aims to achieve and main- tain.
control_mode	It indicates the specific control algorithm or method employed to regulate and maintain the desired pa- rameters within the pipeline.
pump	provides information about the characteristics and behavior of the pump within the pipeline system.
$command_address$	refers to the address associated with a command sent to a certain component within the pipeline system
command_memory	refers to the memory location or data storage area where a command is stored
command_memory_coun	refers to the number of commands or instructions stored in the memory of the control system
comm_read_function	refers to the function or method used for reading or retrieving data from the communication channel or protocol within the control system.
comm_write_fun	refers to the function or method used for writing or sending data to the communication channel or proto- col within the control system.
sub_function	contributes to the overall control and operation of the ICS pipeline.
command_length	It represents the number of characters, bytes, or bits that make up the command.
crc_rate	refers to the CRC (Cyclic Redundancy Check) error rate.
time	refers to the time taken in the opperation
result	refers to the outcome of the particular instance

 Table 3.1: Pipeline dataset Parameters

3.2 Synthetic data generation

3.2.1 Machine Learning Model 1: Generative Adversarial Networks (GANs)

GAN models or Generative Adversarial Network (GAN) model for synthetic data generation. The GAN consists of two neural networks - a generator and a discriminator. The generator creates new synthetic samples, while the discriminator evaluates the quality of the generated samples by comparing them to the real data [32]. There are many classifications within GAN Models, in this case, CTGAN models will be taken into consideration over the normal GAN models for a few reasons.

Both GANs (Generative Adversarial Networks) and CTGAN (Conditional Tabular GANs) are generative models that can be used to create synthetic data that mimics the statistical properties of real data. However, there are some key differences between the two:

- Input data format: GANs can work with any type of data, including images, text, and sound. CTGAN, on the other hand, is designed specifically for tabular data, which is data that is organized into rows and columns.
- Conditioning: CTGAN can generate synthetic data that is conditioned on a set of input features, whereas GANs typically generate data without explicit conditioning. This means that CTGAN can be used to generate synthetic data that is more similar to a specific subset of the real data.
- Training data: CTGAN requires labeled training data, whereas GANs can be trained using unlabeled data. This means that CTGAN is more suitable for supervised learning tasks, where there is a clear distinction between input and output variables.
- Output: GANs generate synthetic data that is optimized to be as realistic as possible, whereas CTGAN generates synthetic data that is optimized to be as similar as possible to the real data, while also satisfying any conditioning constraints.

Overall, CTGAN is a specialized type of GAN that is designed specifically for tabular data, with the ability to generate synthetic data that is conditioned on input features.



Figure 3.2: Generation of synthetic data using the CTGAN model

In this model, we initialize certain parameters before preceding through with the CTGAN model. The dataset, the number of epochs, batch size, the update rate of the discriminator and the generator, and finally the learning rate. The Overall workflow of the CTGAN model follows as per Figure 3.2.

In this model, the data initially undergoes preprocessing in the form of converting categorical columns into one-hot encoded columns and simultaneously another process is also undertaken to normalize the continuous columns. This is done in order to better understand the data and process it using the generators and the discriminator models. Subsequently, the next task at hand is to initialize the generator models and the discriminator models. The generator model is a neural network with multiple fully connected layers that takes a noise vector as input and outputs a synthetic sample in contrast the discriminator model is a neural network with multiple fully connected layers that takes a real or synthetic sample as input and outputs a binary classification (real or fake).

In order to train the discriminator, initially feed a batch of real and synthetic data samples to the discriminator. Once the data has been fed compute the binary cross entropy loss between the true labels (1 for real, 0 for fake) and predicted probabilities. Once that is complete update the discriminator model using back propagation and use the Adam optimizer for accurate and stable results. Repeat the above process for a fixed number of epochs.

Likewise, in order to train the generator, a batch of noise vectors is initially generated. These noise vectors are then subsequently fed to the generator to obtain a batch of synthetic data samples. With the synthetic data samples at hand, feed these samples to the discriminator model in order to obtain the predicted probabilities. Compute the binary cross-entropy loss between the true labels (1 for real, 0 for fake) and predicted probabilities (but swap the labels, i.e., use 1 for fake). Once this process is complete update the generator model using back propagation and the Adam optimizer and repeat the above process for a fixed number of epochs.

Once the generator and the discriminator models have been trained and a batch of resulting synthetic data is obtained, a postprocessing algorithm is applied. This process includes the de-normalization of all the continuous columns and the conversion of the one-hot encoded categorical columns back to categorical columns. With this complete, the resulting output is then evaluated in order to check its performance and likeliness to the original dataset.

With the complete iteration of the CTGAN model, the output obtained is synthetic data that closely resembles the original dataset. This synthetic data are trained in such a manner that it closely resembles the original dataset in terms of statistical similarity and distribution as well [34]. The CTGAN (Conditional Tabular GAN) model offers several benefits in the generation of synthetic data for Industrial Control Systems (ICS) datasets compared to other synthetic data generation models [26]. Some of the key benefits include:

- CTGAN models are used specifically for tabular data. This property makes it suitable for ICS datasets which are in the form of tabular datasets. The CTGAN models unlike the other models are tailored in such a way that it captures the characteristics and dependencies present in the ICS datasets.
- CTGAN models are great at generating synthetic data that possess identical if not close to the original datasets' conditional relationships and dependencies. This is a major factor for ICS datasets as they are mainly comprised of interdependencies within the multitude of variables.

- CTGAN models provide a flexible method by incorporating domain knowledge of conditional variables. This allows for further study that could be done in analyzing the attributes or conditions pertaining to the ICS dataset, therefore, resulting in a context-aware synthetic data generation process.
- CTGAN models employ a large number of deep learning techniques and optimization algorithms to generate synthetic data that is capable of handling large-scale synthetic data This makes it suitable for large ICS datasets in the real world.
- CTGAN models are able to generate highly realistic synthetic data that can mimic the statistical properties and distributions of the original dataset. CT-GAN models, upon learning from the original dataset can replicate the underlying patterns and characteristics present in the synthetic data.
- Evaluation and Validation: CTGAN provides built-in evaluation metrics to assess the quality of the generated synthetic data. These metrics can be used to compare the statistical properties and distributions between the original and synthetic datasets, enabling researchers to validate the effectiveness of the CT-GAN model for generating realistic and reliable synthetic data.

3.2.2 Gaussian Mixture Models (GMM)

The subsequent model in question is the Gaussian Mixture Model. The overall workflow of the model can be seen with the assistance of Figure 3.3.



Figure 3.3: Generation of synthetic data using the GMM model

In this model, the data undergoes a preprocessing step which includes the collection of the ICS dataset and performing a cleaning operation, which consists of removing outliers, handling missing values, and normalizing the data. A feature selection process is also done in order to identify all relevant features from the preprocessed dataset that capture the essential characteristics of the ICS system. The features are then selected based on their significance and relevance in accordance with the specific analysis or modeling objective. This process is predominantly done using correlation and covariance. The selected features are then compiled and fed to the GMM model.

The Gaussian Mixture Model algorithm is then applied to the preprocessed dataset in order to understand and analyze the underlying probability distribution of the data. Once this is complete, the appropriate number of Gaussian components in the GMM is to be determined. This is accomplished using techniques such as the Bayesian Information Criterion (BIC). Once this is complete, estimate the parameters of the GMM, including the means, covariance matrices, and mixture weights, using the expectation-maximization (EM) algorithm.

In order to generate the synthetic dataset, a sampling operation is performed on the GMM model. Using the learned parameters from the above tests synthetic data samples are obtained from the trained GMM model. The parameters that were obtained using the BIC and EM algorithms are used to make the synthetic data resemble the original ICS dataset. The GMM model then generates samples by randomly selecting a component according to the mixture weights and sampling from the corresponding Gaussian distribution.

The synthetic data that is obtained is then adjusted to meet specific requirements or constraints and data transformations or adjustments to align the data with the characteristics of the real ICS datasets are performed. Once the synthetic data is obtained it is imperative to evaluate the quality of the generated synthetic data using various metrics, such as the similarity between the real and synthetic datasets, statistical measures, and domain-specific evaluation criteria.

Once the evaluation is underway a comparison of the statistical properties of the synthetic data with the original ICS dataset to ensure the synthetic data captures the essential characteristics and maintains the integrity of the system. If the quality of the synthetic data does not meet the desired criteria, iterate and refine the GMM model and generation process. Adjust the GMM parameters, such as the number of components, initialization strategy, or covariance structure, and repeat the training and sampling steps. Using the above steps as a roadmap, continuously evaluate the generated synthetic data and refine the process until satisfactory results are achieved.

It is important to note that the benefits of the GMM model may vary depending on the specific characteristics and requirements of the ICS dataset [14]. Other synthetic data generation models, such as Variational Autoencoders (VAEs) or Generative Adversarial Networks (GANs), may also offer unique advantages in certain scenarios. Therefore, the selection of the most suitable synthetic data generation model should consider the specific context, objectives, and characteristics of the ICS dataset under investigation [23][21].

The Gaussian Mixture Model (GMM) has several benefits in the generation of synthetic data for Industrial Control Systems (ICS) datasets compared to other synthetic data generation models [23]. Some of the key benefits include:

- Capturing Complex Data Distributions: GMM is capable of capturing complex data distributions by modeling the data as a mixture of multiple Gaussian components. This allows GMM to represent the underlying structure of the ICS dataset more accurately, especially when the dataset exhibits multi-modal or non-linear distribution patterns.
- Flexibility in Modeling: GMM provides flexibility in modeling the data distribution by allowing the specification of the number of Gaussian components. This flexibility allows the GMM model to adapt to different ICS datasets with varying complexities and capture the inherent variations in the data more effectively.
- Generation of Realistic Samples: GMM can generate synthetic samples that closely resemble the original data distribution. By learning the parameters of the Gaussian components from the real ICS dataset, GMM can produce synthetic data points that preserve the statistical properties, correlation structure, and patterns observed in the original dataset.
- Fine-grained Control over Generated Data: GMM allows fine-grained control over the generated data through the manipulation of the mixture weights and parameters of individual Gaussian components. This control enables researchers to influence the characteristics of the synthetic data, such as adjusting the noise level and skewness level or generating samples with specific characteristics.

- Scalability: GMM can handle large-scale ICS datasets efficiently due to its ability to estimate the parameters using the Expectation-Maximization (EM) algorithm, which is computationally efficient for high-dimensional data. This scalability makes GMM suitable for generating synthetic data for large-scale ICS datasets encountered in real-world industrial systems.
- Interpretable Results: GMM provides interpretable results as it assigns a probability to each data point indicating its association with a particular Gaussian component. This information can be useful in understanding the generated synthetic data and assessing its similarity to the real ICS dataset.

3.2.3 Variational Autoencoders (VAEs)

The final model in consideration in this research thesis is the Variational Autoencoders (VAEs). The overall workflow is shown in Figure 3.4.



Figure 3.4: Generation of synthetic data using the VAE model

In this model the input original ICS dataset is initially preprocessed, i.e. the data is normalized between 0 and 1. This data is then split into training and testing datasets. Once this is complete, the Encoder network is to be initialized.[17]

The encoder network is designed to have an input layer to receive the data, several hidden layers present to understand the representation of the data, and finally, an output layer that represents the latent space z with a mean vector and a standard deviation vector. The encoded data is then fed to the Sampling Layer, where the sample is generated using the reparameterization technique using the mean and the standard deviation vectors.

Subsequently, the decoder network is also initialized, by having an input layer that would receive the sampled data of a particular latent space, several hidden layers to reconstruct the data, and an output layer to generate the synthetic data. There are two different loss functions used in these networks. The First one is the reconstruction loss function which denotes the difference between the input and the output data. And the KL Divergence loss denotes the difference between the learned distribution and the prior distribution.

Once the decoder is then initialized, the training process begins. This involves backpropagating the loss through the model, this is performed in order to update the weights. This process is repeated until the loss function is minimized. Generate the sample of a particular latent space from a standard normal distribution and use the decoder network to generate synthetic data from the sampled latent space. The synthetic data that is obtained from this process is then evaluated using various metrics and the process is then repeated in order to achieve the desired quality.

Overall, the VAE model is an unsupervised learning model that can learn the underlying structure of the input data and generate synthetic data that is similar to the input data [1]. The model can be trained on a variety of input data types, including images, audio, and text. The VAE (Variational Autoencoder) model offers several benefits in the generation of synthetic data for Industrial Control Systems (ICS) [3] datasets compared to other synthetic data generation models. Some of the key benefits include:

- Latent Space Representation: VAEs learn a latent space representation of the data, which allows for the continuous and structured generation of synthetic samples. This means that the VAE can generate data points in a way that smoothly interpolates between different samples, providing more diversity and flexibility in the generated synthetic data.
- Probabilistic Generation: VAEs model the data distribution using probabilistic techniques, allowing for the generation of synthetic data that captures the

inherent uncertainty and variability present in the original ICS dataset. This probabilistic framework enables more realistic and robust synthetic data generation.

- Encoder-Decoder Architecture: VAEs consist of an encoder network that maps the original data into a lower-dimensional latent space, and a decoder network that reconstructs the data from the latent space. This architecture enables the VAE to capture the complex dependencies and patterns present in the ICS dataset, resulting in more accurate and realistic synthetic data generation.
- Continuous and Disentangled Latent Representations: VAEs encourage disentangled representations in the latent space, meaning that different dimensions of the latent space capture independent and meaningful factors of variation in the data. This property allows for better control over the generation process, as specific features or attributes can be manipulated in the latent space to generate synthetic samples with desired characteristics.
- Ability to Learn Complex Distributions: VAEs are capable of learning complex data distributions, including multimodal distributions, which are often encountered in ICS datasets. This capability ensures that the generated synthetic data captures the diversity and variability of the original dataset, leading to more accurate representations of real-world scenarios.
- Reconstruction-Based Evaluation: VAEs can be evaluated based on the quality of the reconstructed data. By comparing the original and reconstructed samples, researchers can assess the fidelity and accuracy of the VAE model in capturing the characteristics of the ICS dataset. This evaluation provides insights into the reliability and effectiveness of the VAE for synthetic data generation.
- Interpretable Latent Space: VAEs can provide interpretable representations in the latent space, where each dimension corresponds to a meaningful attribute or feature of the data. This interpretability facilitates better understanding and analysis of the generated synthetic data, as researchers can examine and manipulate specific dimensions to explore different aspects of the data distribution.

With the generation of synthetic data using the three machine learning algorithms namely CTGAN, GMM, and VAE models complete and usable synthetic data has been generated, and a thorough evaluation process will now commence.

Chapter 4

Evaluation of Synthetic Data

One of the main objectives of this thesis is to establish a structured evaluation pattern by which one could formulate and understand the statistical patterns and trends along with an extensive evaluation of newly generated synthetic data and how well it performs when compared with the original ICS datasets. Along with this, it is also imperative to provide a comparative analysis of the different means of generating synthetic data and displaying their characteristics using various means, namely using Statistical analytics, Visual analytics, and Machine learning algorithms to provide an in-depth view of the synthetic data.

Evaluating the quality of synthetic data is crucial to ensure its reliability, usefulness, and applicability in various domains and applications. When assessing the quality of synthetic data, it is essential to consider multiple categories or criteria. These categories provide a comprehensive framework for evaluating different aspects of synthetic data and enable researchers and practitioners to make informed decisions. In this section, we will discuss the key categories that one must check to evaluate the quality of synthetic data.

- Fidelity: Fidelity refers to the degree to which the synthetic data accurately represents the original dataset [24]. It involves assessing how well the statistical properties, distributions, and relationships of the original data are preserved in the synthetic data. Measures such as mean, variance, covariance, and higher-order statistics can be used to compare the statistical properties between the original and synthetic datasets. Additionally, visual inspection, hypothesis testing, and domain experts' judgment can provide insights into the fidelity of the synthetic data.
- Privacy and Information Preservation: Privacy preservation is a critical consideration when dealing with synthetic data, especially when the original data

contains sensitive or personal information. Evaluating privacy preservation involves assessing how well the synthetic data obscures or anonymizes personal attributes or sensitive data [35]. Metrics such as k-anonymity, l-diversity, and t-closeness can be used to measure the level of privacy protection provided by synthetic data. Additionally, techniques such as differential privacy can be applied to ensure privacy guarantees. Using these methods we can also check if sensitive and vital information is preserved in the newly generated synthetic data.

- Diversity and Generalization: Diversity measures the variety and coverage of the synthetic data across different attributes and feature combinations. It assesses whether the synthetic data captures the inherent variability present in the original dataset. Metrics such as entropy, distinctiveness, or clustering techniques can be used to measure the diversity of the synthetic data [36]. A diverse synthetic dataset ensures that it represents the full range of patterns and characteristics present in the original data. Generalization assesses how well the synthetic data generalize to unseen or out-of-sample data. It measures the robustness of the synthetic data generation method in capturing the underlying patterns and characteristics of the original dataset. It is important to ensure that the synthetic data accurately represents the real-world distribution of the data to ensure reliable and robust generalization.
- Interpretability and Utility: Interpretability refers to the understandability and transparency of the synthetic data generation process. It involves assessing whether the synthetic data can be easily interpreted and analyzed by domain experts [37]. Techniques such as feature importance ranking, feature engineering, or model interpretability methods can be applied to enhance the interpretability of synthetic data. Utility refers to the usefulness and applicability of synthetic data in specific tasks or applications. It involves assessing how well the synthetic data performs in downstream tasks, such as predictive modeling, classification, or clustering. The performance of models trained on synthetic data to measure the utility of the synthetic data.

In this chapter, we will be exploring these subtopics in detail in order to determine and understand their usage and explore the results of the synthetic data generated by the various machine learning model namely the CTGAN, GMM, and VAE models.

4.1 Fidelity

Fidelity is a critical aspect when evaluating the quality of synthetic data. It refers to the extent to which the synthetic data accurately represents the statistical properties, distributions, and relationships present in the original dataset. In other words, fidelity measures how well the synthetic data captures the essence of the original data [24].

To assess fidelity, various statistical measures and techniques can be employed. Here are some key considerations when evaluating fidelity in synthetic data:

- Statistical Properties: Fidelity evaluation involves comparing the statistical properties of the original and synthetic datasets. This includes measures such as mean, variance, covariance, skewness, and kurtosis. Statistical tests, such as t-tests, can be applied to determine if there are significant differences between the statistical properties of the original and synthetic data [24]. A high degree of similarity in statistical properties indicates a higher fidelity of the synthetic data.
- Distribution Matching: Synthetic data should ideally replicate the distributions observed in the original data. This includes capturing the shape, spread, and tail behavior of the distributions. Visual inspection, kernel density estimation, or quantile-quantile plots can be used to compare the distributions of original and synthetic data [24]. Close alignment between the distributions indicates a higher fidelity of the synthetic data.

It is important to note that achieving perfect fidelity between the synthetic and original datasets may not always be feasible or necessary. The goal is to strike a balance between fidelity and privacy protection. Synthetic data may intentionally introduce some level of noise or perturbation to preserve privacy, which may result in slight deviations from the original data. The level of fidelity required will depend on the specific application and the trade-off between privacy and utility. Overall, fidelity assessment provides insights into the accuracy and reliability of the synthetic data. By ensuring a high level of fidelity, researchers and practitioners can have confidence in using synthetic data for various analyses, modeling, and decision-making processes.

4.1.1 Descriptive Statistics

One of the primary and important techniques that could be used to determine the statistical properties is descriptive statistics. Descriptive statistics take into account the mean, mode, range, variance, and standard deviation [12][29]. Using this it is possible to determine whether the statistical properties possessed by the synthetic data are close to the original dataset.

Table 4.1: Descriptive statistics output for original dataset

Para	res_add	resp_lgt	res_mem	res_wrt_fun	res_mem_cn	t res_rd_fun	setpoint	ctrl_md	pump	result
Count	97019	97019	97019	97019	97019	97019	97019	97019	97019	97019
Mean	3.719	26.29	216.65	9.298	16.737	2.44	24.16	0.899	0.056	0.36
Std	1.021	26.56	59.5	2.55	4.59	0.893	14.322	0.991	0.23	0.482

Table 4.2: Descriptive statistics output for CTGAN dataset

Para	res_add	resp_lgt	res_mem	res_wrt_fun	res_mem_cn	t res_rd_fun	setpoint	ctrl_md	pump	result
Count	97019	97019	97019	97019	97019	97019	97019	97019	97019	97019
Mean	1.14	48.77	66.48	2.88	5.15	1.98	39.77	0.97	0.953	0.053
Std	0.512	12.12	27.07	1.19	2.11	0.193	8.101	0.17	0.211	0.224

Table 4.3: Descriptive statistics output for GMM dataset

Para	res_add	resp_lgt	res_mem	res_wrt_fun	res_mem_cn	t res_rd_fun	setpoint	ctrl_md	pump	result
Count	97019	97019	97019	97019	97019	97019	97019	97019	97019	97019
Mean	3.714	26.29	216.4	9.286	16.71	2.34	24.26	0.899	0.033	0.37
Std	1.029	26.77	59.99	2.574	4.63	0.903	14.509	1.026	0.238	0.483

Table 4.4 :	Descriptive	statistics	output fo	r VAE	dataset
---------------	-------------	------------	-----------	-------	---------

Para	res_add	resp_lgt	res_mem	res_wrt_fun	res_mem_cn	t res_rd_fun	setpoint	ctrl_md	pump	result
Count	97019	97019	97019	97019	97019	97019	97019	97019	97019	97019
Mean	3.196	40.73	185.2	7.621	14.24	2.649	34.318	0.589	0.0047	0.0053
Std	0.652	14.38	32.32	1.382	2.565	0.490	9.645	0.539	0.211	0.2249

The results obtained from the descriptive analysis are displayed in the tables above, representing the statistical quality and representation of the synthetic dataset. These tables are interpreted as follows: If the statistical parameters generated by the synthetic data generation model differ significantly from those of the original ICS dataset, the synthetic data generation model in question is deemed unsuitable and can be eliminated from further testing. On the other hand, if the synthetic data generation model can generate these statistical parameters with an acceptable level of similarity to the original ICS dataset, it can be considered a suitable model.

4.1.2 Mahalanobis Distance

From [12] Mahalanobis distance is a statistical measure that quantifies the distance between a point and a distribution of points in a multivariate space. It considers the correlations between variables, which makes it particularly useful when dealing with datasets with multiple variables that are not necessarily independent. The Mahalanobis distance is a normalized measure that considers the variances and covariances of the variables in the dataset. It is essential in evaluating the similarity between an observation and the distribution it belongs to. This test allows one to check the authenticity of the synthetic dataset. Using this test it can be found whether the dataset is from the same distribution or from a different one and also how different the dataset is when compared with the original dataset.

The Mahalanobis distance calculation can be divided into three main steps. Initially, the data is preprocessed by centering and scaling it. This can be done by subtracting the mean value from each variable and further dividing it by the standard deviation. This method ensures that all variables have been scaled and there are no overrepresentations or dominance by variables with larger value ranges. The next step is the calculation of the covariance matrix in order to capture the relationships and correlations between variables. If there are n variables, the covariance matrix will be an n x n matrix. Once the covariance matrix is obtained, the Mahalanobis distance can be calculated for a given observation. The formula for calculating the Mahalanobis distance is:

Mahalanobis Distance = $\sqrt{(\mathbf{x} - \boldsymbol{\mu})^{\top} \boldsymbol{\Sigma}^{-1} (\mathbf{x} - \boldsymbol{\mu})}$

x represents the vector of variables for the observation being evaluated. μ represents the vector of means of the reference distribution. $-\sum(-1)$ represents the inverse of the covariance matrix of the reference distribution.

In simpler terms, the Mahalanobis distance is the Euclidean distance between the centered and scaled observation and the centered and scaled reference distribution, where the scaling is done using the covariance matrix [29].

Model	Mahalanobis Distance Mean	Standard deviation
Original	71.6008	61.567
GAN	43.598	28.625
GMM	72.606	62.353
VAE	59.472	43.776

Table 4.5: Mahalanobis distance mean Comparison.

From the table above, it is inferred that the smaller the difference between the original dataset mean and the chosen synthetic data generation model, the greater the similarity between the two datasets. The tables demonstrate that the GMM model has the greatest similarity with the original dataset, followed by the VAE model and the GAN model.

4.1.3 Hotelling T2 test

The Hotelling T2 test, also known as Hotelling's T-squared test, is a multivariate statistical test used to determine if there is a significant difference between the means of two groups in a multivariate setting. It is an extension of the univariate two-sample t-test to the multivariate case. In the multivariate setting, we have multiple variables or features measured on each observation. The Hotelling T2 test takes into account the covariance structure between these variables when comparing the means of two groups.

The test evaluates whether the mean vectors of the two groups are significantly different from each other in the multivariate space. It considers both the location and spread of the data in multiple dimensions. The test calculates a T2 statistic, which is a measure of the distance between the two mean vectors, taking into account the covariance structure [22][25][28]. The null hypothesis of the Hotelling T2 test is that there is no difference between the means of the two groups. The alternative hypothesis is that there is a significant difference between the means. The Hotelling T2 test requires certain assumptions to be valid, including multivariate normality and homogeneity of covariance matrices. Violations of these assumptions can affect the accuracy of the test results.

The test statistic follows Hotelling's T-squared distribution under the null hypothesis. Based on the calculated T2 statistic, along with the sample sizes and the number of variables, p-values can be obtained to determine the significance of the test. The Hotelling T2 test can also be applied to evaluate the quality and similarity of synthetic data compared to the original data. In the context of synthetic data evaluation, the Hotelling T2 test assesses whether the mean vectors of the synthetic data and the original data are significantly different from each other in a multivariate space. When evaluating synthetic data, the goal is to ensure that the synthetic dataset captures the statistical properties and characteristics of the original dataset. The Hotelling T2 test can help determine if the synthetic data closely approximates the distribution of the original data by comparing their mean vectors.

The procedure for applying the Hotelling T2 test to synthetic data evaluation involves the following steps. Prepare the synthetic dataset and the original dataset, ensuring that they have the same number of variables or features. Calculate the mean vector for both the synthetic data and the original data by taking the average values of each variable across the respective datasets. Estimate the covariance matrices for both datasets. The covariance matrix describes the relationships and dependencies between the variables in the data. Compute the Hotelling T2 test statistic using the mean vectors and covariance matrices of the synthetic and original datasets. The test statistic quantifies the difference between the mean vectors, accounting for the covariance structure. Evaluate the significance of the test statistic by comparing it to the critical values from Hotelling's T-squared distribution. Calculate the pvalue associated with the test statistic to determine whether the mean vectors of the synthetic and original datasets are significantly different. If the p-value is below a predetermined significance level (e.g., 0.05), it suggests that there is a significant difference between the synthetic and original datasets. On the other hand, a higher pvalue indicates that the synthetic data closely resembles the original data in terms of their mean vectors. By applying the Hotelling T2 test, researchers and practitioners can quantitatively assess the quality and similarity of synthetic data to the original data. This evaluation method provides an objective measure of the extent to which the synthetic data captures the statistical characteristics of the original data.

Model	T2 test p-value
GAN	0.012
GMM	1 11e-16
	1.110 10

Table 4.6: Hotelling T2 test p-value Comparison

Similar to the Mahalanobis distance, the similarity between the synthetic dataset in question and the original dataset is taken into consideration. As the p-value approaches 0, it can be seen that the datasets are from different distributions but possess similar statistical similarities. From the table above, it is evident that the GMM model performs the best, followed by the VAE model and the GAN model.

4.2 Privacy and Information Preservation

Privacy and information preservation are crucial considerations when evaluating synthetic data. The evaluation process should ensure that the generated synthetic data preserves the privacy of individuals and does not disclose sensitive or personally identifiable information. Information preservation is an essential aspect of synthetic data evaluation, ensuring that the synthesized data retains the important statistical and structural properties of the original dataset [6]. The goal is to generate synthetic data that captures the essential characteristics and patterns of the original data while protecting sensitive information. Here are some techniques and considerations for information preservation in synthetic data evaluation:

• Statistical Properties: The synthesized data should maintain key statistical

properties of the original dataset, such as distributions, correlations, and summary statistics. This ensures that the synthetic data accurately represents the underlying patterns and relationships present in the original data.

- Feature Preservation: It is important to preserve the essential features of the original dataset in the synthetic data. Features that are highly relevant to the analysis or modeling tasks should be accurately represented to maintain the data's utility and usefulness.
- Structural Integrity: The synthetic data should preserve the structural integrity of the original dataset. This involves maintaining the overall data structure, including hierarchical relationships, dependencies, or any other structural characteristics that are important for the analysis.
- Data Completeness: The synthetic data should adequately cover the range of values and patterns present in the original dataset. It should accurately represent the data's diversity and variability to ensure comprehensive analysis and modeling.

By focusing on information preservation during synthetic data evaluation, researchers can ensure that the generated data maintains its integrity, relevance, and usefulness. This facilitates reliable analysis, modeling, and decision-making processes while protecting sensitive information and complying with privacy regulations.

4.2.1 Mutual Information Score (MI)

Mutual Information (MI) is a statistical measure used in synthetic data evaluation to assess the amount of information shared between two variables or datasets. It quantifies the degree of dependency or association between variables, indicating how much knowledge of one variable can provide insights into the other.

In the context of synthetic data evaluation, MI is used to compare the information content of the original dataset and the synthetic dataset. It helps to determine how well the synthetic data captures the information present in the original data. A higher MI score indicates a stronger relationship and greater similarity between the variables or datasets being compared [27]. The MI score is based on the concept of entropy, which measures the uncertainty or randomness of a variable. By calculating the entropy of the original data and the synthetic data separately and then comparing their joint entropy, the MI score can be derived. The MI score can range from 0 to a maximum value, where 0 indicates no relationship or information shared, and the maximum value represents a perfect relationship.

In synthetic data evaluation, the MI score can be used to assess the fidelity and information preservation of the synthetic data. A higher MI score suggests that the synthetic data closely resembles the original data in terms of the underlying patterns, dependencies, and information content. On the other hand, a lower MI score indicates a potential loss of information or deviation from the original data's characteristics.

Let's consider two variables X and Y, where X represents the original dataset and Y represents the synthetic dataset.

Calculate the individual entropies of X and Y:

Entropy of $X : H(X) = -\sum p(x) \log p(x)$, where p(x) is the probability distribution of X. Entropy of $Y : H(Y) = -\sum p(y) \log p(y)$, where p(y) is the probability distribution of Y. Calculate the joint entropy of X and Y:

Joint Entropy of X and Y: $H(X, Y) = -\sum \sum p(x, y) \log p(x, y)$ is the joint probability distribution of X and Y. Compute the MI score using the formula:

MI(X, Y) = H(X) + H(Y) - H(X, Y) The MI score represents the reduction in uncertainty or the amount of information gained about one variable by knowing the other variable. A higher MI score indicates a stronger relationship or more shared information between X and Y.

It's important to note that the estimation of probabilities and the calculation of entropy and joint entropy may vary depending on the specific data and the chosen estimation method. Different techniques such as histogram-based estimation, kernel density estimation, or nearest-neighbor methods can be employed to estimate the probability distributions.

By comparing the MI score between the original dataset (X) and the synthetic dataset (Y), researchers can assess the similarity in terms of the shared information and evaluate the fidelity and information preservation of the synthetic data [27][31].

In this test we take the MI score difference ie. |MI(synthetic) – MI(Original)|. The

absolute value derived from this expression denotes the level of information retention that is present between the original dataset and the synthetic dataset generated by the three different models The scores obtained for the respective models are as follows.

Model	MI Score abs	MI Score synthetic
Original		0.99
GAN	0.77	0.22
GMM	0.109	0.881
VAE	0.44	0.55

Table 4.7: MI Scores of each model

This test is a good indicator to check the similarity between the distributions. The lower the MI score, the greater the information captured by the synthetic data. From the table above, it can be observed that the GMM model has the lowest MI score compared to the other synthetic datasets. This is followed by the VAE model and the GAN model.

4.2.2 Kernel Density Estimation (KDE)

Kernel Density Estimation (KDE) is a non-parametric method used to estimate the probability density function (PDF) of a random variable based on a given set of observations. It is commonly employed in synthetic data evaluation to assess the distributional similarity between the synthetic dataset and the original dataset [4].

The basic idea behind KDE is to represent the PDF as a weighted sum of kernel functions centered at each observation. The kernel function is a smooth, symmetric, and non-negative function that determines the shape of the estimated density

The KDE process involves the following steps. The choice of kernel function influences the smoothness and accuracy of the estimated density. The Gaussian kernel is a popular choice due to its smoothness and mathematical properties. The bandwidth parameter controls the width of the kernel function and affects the smoothness of the estimated density. A small bandwidth leads to a more detailed density estimate but may overfit the data, while a large bandwidth results in a smoother estimate but may over smooth the data. Selecting an appropriate bandwidth is crucial to obtain an accurate density estimate. For each observation in the dataset, the kernel function is centered at that point, and the density contribution is calculated. The contributions from all observations are summed to obtain the overall density estimate. The resulting estimated density function represents an approximation of the underlying PDF. It can be evaluated at any point to obtain the density value.

In the context of synthetic data evaluation, KDE can be used to estimate the PDF of the original dataset and compare it with the PDF of the synthetic dataset. By comparing the two density estimates, researchers can assess the similarity in terms of shape, peaks, and overall distributional characteristics. If the synthetic data closely matches the original data's density, it indicates a higher level of fidelity and preserves the statistical properties of the original dataset.



Figure 4.1: KDE graph for GAN models, comparing features response_address (Left) and resp_length (right).



Figure 4.2: KDE graph for GMM models, comparing features response_address (left) and resp_length (right).



Figure 4.3: KDE graph for VAE models, comparing features response_address and resp_length (right).

4.2.3 Wasserstein distance

The Wasserstein distance, also known as the Earth Mover's distance (EMD), is a metric used to quantify the difference between two probability distributions. It is often employed in synthetic data evaluation to assess the similarity between the probability distributions of the original and synthetic datasets [29][33].

Unlike other distance metrics that focus on comparing individual data points, the Wasserstein distance considers the entire distribution and measures the minimum cost of transforming one distribution into another. It accounts for the underlying structure and shape of the distributions, making it suitable for capturing differences in their probability density functions (PDFs).

The basic concept behind the Wasserstein distance involves considering the distributions as "piles of earth" and measuring the minimum amount of "work" required to transform one pile into the other. The work is calculated as the product of the amount of earth moved and the distance it is moved. This distance can be defined based on various metrics, such as the Euclidean distance or the Minkowski distance.

However, computing the Wasserstein distance can be computationally expensive, especially for high-dimensional data or large datasets. Various algorithms, such as the Kantorovich-Rubinstein duality or Sinkhorn algorithm, have been developed to efficiently estimate the Wasserstein distance for practical applications.

In synthetic data evaluation, the Wasserstein distance can be used to compare the distributions of individual variables or the joint distribution of multiple variables between the original and synthetic datasets. By quantifying the dissimilarity, researchers can assess the quality and fidelity of the synthetic data generation process. The mathematical working of the Wasserstein distance involves finding the optimal transportation plan that minimizes the cost of transforming one distribution into another. It considers the amount of mass being transported and the distance it needs to be moved. The distance metric used to measure the transportation cost can vary, but the most commonly used is the Euclidean distance.

Let's consider two probability distributions, P and Q, with probability density functions (PDFs) p(x) and q(x) respectively. The Wasserstein distance between P and Q, denoted as W(P, Q), can be calculated using the following formula:

 $W(P,Q) = \inf\left(\int \int \|x - y\| \cdot T(x,y) \, dx \, dy\right)$

where the infimum is taken over all possible transportation plans T(x, y) that satisfy the following constraints:

Marginal constraints: The total amount of mass transported from each point in P must equal the corresponding point in Q. This ensures that no mass is lost or gained during the transportation process.

$$\int T(x, y) dy = p(x) \text{ for all } x$$
$$\int T(x, y) dx = q(y) \text{ for all } y$$

Non-negativity constraints: The transportation plan T(x, y) should be non-negative, meaning that it cannot transport negative mass.

$$T(x,y) \ge 0$$
 for all x, y

The Wasserstein distance measures the minimum cost of transporting mass from P to Q, where the cost is determined by the distance ||x-y|| multiplied by the amount of mass being transported T(x, y). By finding the optimal transportation plan that minimizes this cost, the Wasserstein distance provides a measure of dissimilarity between the two distributions [29][8].

Using this as the base the Wasserstein distance is measured between the original dataset and the synthetic dataset. The results obtained are as follows: measure the distance between two probability distributions

A low Wasserstein distance between the original and synthetic data indicates that the synthetic data captures the same distribution as the original data.

Model	Wasserstein distance
GAN	0.534
GMM	0.106

Table 4.8: Wasserstein distance Scores of each model

This test provides us with a method to compare the probability distributions of the two datasets. Therefore, the closer the distance between the two datasets, the greater is the similarity between distributions and the relationship between each feature. From the above table, it can be noted that the GMM model displays better results than the VAE model and the GAN model.

4.2.4 t-SNE plot visualizations

t-SNE (t-Distributed Stochastic Neighbor Embedding) is a dimensionality reduction technique commonly used for visualizing high-dimensional data in a lower-dimensional space. It is often employed in synthetic data evaluation to examine the structure and clustering patterns of the synthetic data in comparison to the original data [28].

The working of t-SNE involves transforming the high-dimensional data into a two- or three-dimensional representation while preserving the local similarities between data points. It achieves this by modeling pairwise similarities using probability distributions. The algorithm starts by computing pairwise similarities between the high-dimensional data points using a Gaussian kernel. The similarities are then converted into probabilities using a Student's t-distribution.

The t-SNE algorithm iteratively maps the high-dimensional data points to the lowdimensional space, aiming to minimize the divergence between the pairwise similarity distributions of the high-dimensional data and the low-dimensional embeddings. The mapping process is performed by optimizing the Kullback-Leibler (KL) divergence between the two distributions.

This is a powerful visualization technique that enables the exploration and comparison of high-dimensional data in a lower-dimensional space. It provides a visual representation of the structures and clustering patterns of the data. This can assist in determining the similarity and quality of synthetic data when compared with the original data [28].



Figure 4.4: Cluster Identification using t-SNE plots

4.3 Diversity and Generalization

Evaluating the diversity of synthetic data is an essential aspect of assessing the quality and usefulness of generated synthetic datasets. Diversity refers to the extent to which the synthetic data captures the variability and distribution of the original data. A diverse synthetic dataset should accurately represent the different patterns, relationships, and characteristics present in the original dataset [29].

Evaluating diversity is crucial because a lack of diversity in synthetic data can lead to biased or incomplete representations, which may limit the effectiveness and applicability of the generated data. Therefore, it is important to employ evaluation measures and techniques that quantitatively assess the diversity of the synthetic data.

There are several approaches to evaluating the diversity of synthetic data. One common method is to compare statistical properties between the original and synthetic datasets. This can include analyzing summary statistics, distributional characteristics, or correlations of relevant variables. If the synthetic data closely matches these statistical properties, it indicates a higher level of Diversity [30].

Additionally, visualization techniques can be employed to visually inspect the diversity of the synthetic data. Scatter plots, histograms, or t-SNE plots can help visualize the distribution and clustering patterns of the synthetic data in comparison to the original data. Visual assessments can provide valuable insights into the diversity of the synthetic data and help identify any discrepancies or limitations.

4.3.1 Histograms and Scatter Plots

In order to compare and visualize the distributions of particular features in the original and synthetic datasets Histograms prove to be one of the most useful tools. Taking a particular feature into consideration that belongs to both the original dataset and the synthetic dataset when they are plotted, it becomes easier to identify and analyze discrepancies or similarities between the two datasets.

Using this method, if the features of the synthetic dataset closely match that of the original dataset, then it is likely that the synthetic dataset has a good representation of the original data. Likewise, if there are a lot of discrepancies between the two features on the histogram then the synthetic data does not capture the distribution of the features correctly [30].

Histograms can also be used to identify any outliers in the synthetic data. If there are large spikes or gaps in the histogram of the synthetic data, it could indicate the presence of outliers that need to be investigated further.

Overall, histograms provide a quick and easy way to compare the distribution of a feature between the original and synthetic data and can be a useful tool in synthetic data analytics.



Figure 4.5: Feature Distribution graph, Comparing Original and Synthetic datasets (GMM)

Similar to the histograms scatterplots are a useful visualization tool for synthetic data analytics. They are used to plot the relationship between two variables in a



Figure 4.6: Feature Distribution graphs using Histograms



Figure 4.7: Feature Distribution graphs using Scatter Plots

dataset. In the context of synthetic data analytics, scatterplots are used to visually compare the distribution of variables in the original and synthetic datasets [7].

If the scatterplot shows a linear relationship between the variables in both datasets, it suggests that the synthetic data is a good representation of the original data. However, if the scatterplot shows a nonlinear relationship or there are significant differences between the variables in the two datasets, it may indicate that the synthetic data needs further refinement.

In addition to comparing the distribution of variables, scatterplots can also be used to identify outliers in the synthetic dataset. Outliers are data points that deviate significantly from the overall pattern of the data. If the scatterplot shows outliers in the synthetic dataset that were not present in the original data, it may indicate that the synthetic data is not representative of the original data and needs to be refined [38].

4.3.2 Kolmogorov-Smirnov (KS) test

The Kolmogorov-Smirnov test is a statistical test used to evaluate the similarity between two probability distributions. In the context of synthetic data evaluation, the Kolmogorov-Smirnov test can be employed to assess the similarity between the distribution of the original dataset and the distribution of the synthetic dataset [27].

The test works by comparing the empirical cumulative distribution functions (CDFs) of the two datasets. The empirical CDF represents the proportion of data points in a dataset that are less than or equal to a given value. The Kolmogorov-Smirnov test calculates the maximum absolute difference (D) between the two empirical CDFs. This difference, also known as the Kolmogorov-Smirnov statistic, serves as a measure of dissimilarity between the distributions.

To perform the Kolmogorov-Smirnov test, the null hypothesis is that the two datasets are drawn from the same distribution. If the calculated Kolmogorov-Smirnov statistic exceeds a critical value (typically obtained from statistical tables), the null hypothesis is rejected, indicating that the two distributions significantly differ from each other [26].

In the context of synthetic data evaluation, the Kolmogorov-Smirnov test can be used to determine whether the distribution of the synthetic dataset matches that of the original dataset. A smaller Kolmogorov-Smirnov statistic suggests a higher similarity between the distributions, indicating that the synthetic data is more representative of the original data. The results obtained ie. the corresponding p-value to each synthetic data generation model is as follows:

 Table 4.9: Kolmogorov-Smirnov (KS) test Scores of each model

Model	Kolmogorov-Smirnov test p-value
GAN	0.12
GMM	3.1175 e-29

The KS test provides vital information for checking whether the two datasets in consideration are part of a larger dataset. This test acts as a quality control metric to prevent future machine learning model classifiers from being tricked by synthetic data. It also allows us to analyze the similarity in distribution patterns. From the results obtained, it can be inferred that the GMM model possesses the greatest result, followed by the VAE model and the GAN model, respectively.

4.4 Interpretability and Utility

Evaluating the model interpretability and utility of synthetic data is crucial to assess the effectiveness and practicality of using synthetic data in various applications. Model interpretability refers to the ability to understand and interpret the inner workings and decisions made by a model, while utility refers to the usefulness and effectiveness of the synthetic data in achieving the intended goals [12][31].

In the context of synthetic data evaluation, model interpretability involves assessing how well the synthetic data captures the underlying patterns, relationships, and features present in the original data. It requires evaluating whether the synthetic data retains the important characteristics of the original data, such as the distributional properties, correlations between variables, and relevant patterns or trends. Various techniques can be employed to assess interpretability, including visualizations, statistical analysis, and comparison with domain knowledge.

On the other hand, evaluating the utility of synthetic data focuses on determining the extent to which the synthetic data can effectively substitute the original data in practical applications or analyses. This involves evaluating the performance of models or algorithms trained on the synthetic data and assessing their ability to achieve the desired outcomes. Utility evaluation may involve comparing the performance of models trained on the synthetic data with models trained on the original data, considering metrics such as accuracy, precision, recall, or any other relevant performance measure [23].

4.4.1 Decision Trees and Feature Importance

Evaluating the model interpretability of synthetic data is essential to understand the underlying patterns and decision-making processes of the generated data. One popular approach for assessing interpretability is through the use of decision trees.

Decision trees are hierarchical models that make decisions based on a sequence of rules or conditions. They provide a transparent and intuitive representation of how the input variables influence the output or target variable. When applied to synthetic data evaluation, decision trees can reveal the important features and relationships within the data, helping to understand how the synthetic data captures the patterns present in the original data [13].

The use of decision trees for the model interpretability of synthetic data offers several advantages. Decision trees provide a clear and transparent representation of the decision-making process, making it easier to understand and interpret the generated data. They also allow for the identification of important features and their relative importance, aiding in feature selection and dimensionality reduction. Furthermore, decision trees can handle both numerical and categorical variables, making them suitable for various types of data.

Accuracy for Classificatio	original data n report for	: 0.9304; original	78251906823 data:	3	Accuracy for of Classification	original data n report for	: 0.93047 original	8251906823: data:	3	Accuracy for o	original data n report for	: 0.93047 original	78251906823 data:	3
				support										
0	0.92	0.98	0.95	12284			0.98	0.95	12284			0.98	0.95	
1	0.96	0.85	0.90			0.96	0.85	0.90				0.85	0.90	
accuracy			0.93	19404				0.93	19404				0.93	19484
macro avg	0.94	0.91	0.92	19404	macro avg	0.94	0.91	0.92	19404	macro avg	0.94	0.91	0.92	19404
weighted avg	0.93			19404	weighted avg	0.93	0.93	0.93	19404	weighted avg	0.93	0.93	0.93	19404
Accuracy for			12279942279	94			a: 0.9232	11708925994						
Classificatio	n report for		: data:									synthetic	: data:	
				support									f1-score	
	0 00	0 00	0 00	12284			0.98							
1	0.32	0.82	0.46	7120			0.82	0.89				0.01	0.02	
ACCUMPACIA			0.30	10404	accuracy			0.92	19404				0.64	19484
macco ave	8 16	9 41	0.30	19404	macro avg		0.90	0.91	19404		0.82	0.51	0.40	19404
weighted avg	0.12	0.30	0.17	19404	weighted avg		0.92	0.92	19404	weighted avg	0.77	0.64	0.50	19404
G	GAN Mo	odel			GN	ИМ Мо	del			۷	AE Mo	del		

Figure 4.8: Performance Matrix of Synthetic data using Decision Tree's

The second section of this test is performed to check if the original dataset and the synthetic data are from the same distribution or not. This is done by merging the two datasets and adding a target column to indicate if the corresponding instance is from the original dataset or the synthetic dataset. The model's capability to detect if the instance is real or synthetic can determine if the dataset has been spliced from one another ie. the lower the prediction accuracy greater the chance of datasets being generated from different distributions [12]. The results are as follows:

This test is known as the "adversarial test". In this test, the synthetic data is evaluated using a pre-trained classifier, and the objective is to see how well the

Model	Accuracy of the Decision tree
GAN	72%
GMM	46%
VAE	58%

Table 4.10: Prediction based accuracy rate of each model

classifier can distinguish between the synthetic and real data.

The significance of this test lies in its ability to measure the quality and realism of the synthetic data. A good synthetic dataset should be indistinguishable from real data, meaning that the classifier should perform poorly in distinguishing between the two. If the classifier can easily differentiate between the synthetic and real data, it indicates that the synthetic data lacks the necessary characteristics and statistical properties of real data.

Having a lower accuracy (or higher misclassification rate) in the adversarial test is actually desirable in this case. It means that the synthetic data is more realistic and representative of the real data, as the classifier is finding it difficult to distinguish between them. A lower accuracy suggests that the synthetic data is successfully capturing the underlying patterns and statistical properties of the real data, making it a useful and high-quality substitute for real data.

Therefore from the above table it can be inferred that the GMM model has the greatest performance when compared to the VAE model and the GAN model respectively.

The third section of this test is performed in order to determine the feature importance of each feature in the original dataset and the synthetic dataset and compare them. Feature importance analysis provides insights into the relevance and contribution of individual input variables in the model's decision-making process [12].

The process of evaluating model interpretability using feature importance typically involves the following steps. The synthetic data is used to train a machine learning model, such as a decision tree, random forest, or gradient boosting model. These models are capable of capturing complex relationships between the input variables and the target variable. After training the model, the feature importance is calculated based on the model's internal mechanisms. Various techniques can be employed to measure feature importance. In linear models or models with interpretable coefficients (e.g., logistic regression), the magnitude of the coefficients can be used to assess feature importance. Larger coefficients indicate greater importance.

Evaluating model interpretability using feature importance provides several benefits. It allows researchers to understand the factors influencing the model's predictions and identify the most influential features in the synthetic data. This knowledge can help detect potential biases, assess the generalizability of the synthetic data, and guide further data generation or refinement efforts [12][31].

GAN Model:

Original Feature Importance: [0.04047268 0.03657814 0.04702534 0.03535199 0.03022612 0.29781821 0.23147441 0.26938562 0.01166749] Synthetic Feature Importance: [0.00473554 0.18249142 0.44117087 0.01748763 0.07397905 0.00145167 0.17235447 0.00100415 0.10532521]

GMM Model:

Original Feature Importance: [0.04342409 0.04388343 0.03370296 0.02628716 0.04178168 0.31432631 0.23506253 0.25148647 0.01004537] Synthetic Feature Importance: [0.03338723 0.02646236 0.02719719 0.03393762 0.0384217 0.24115417 0.35221881 0.23982815 0.00739278]

VAE Model:

Original Feature Importance: [0.03215307 0.0295622 0.03265624 0.04924049 0.04402649 0.3202203 0.2404422 0.23870251 0.01299649] Synthetic Feature Importance: [0.00557989 0.14308656 0.27891049 0.0118397 0.02963738 0.04492673 0.44546204 0.02268573 0.01787148]

Figure 4.9: Feature importance with reference to model coefficients.

4.5 Comparative Analysis

Using the results obtained in the above section, a comparative analysis report is formed. This section of the thesis contains a comprehensive analysis of the results obtained above and revisiting and explaining the significance of each subsection. The evaluation is divided into four subsections as per the suggested evaluation framework under the subheadings of Fidelity, Privacy and Information Preservation, Diversity and Generalization, and finally Interpretability and Utility. Through the above subsections, each topic is explored along with the corresponding test being either statistics, visualization, or machine learning oriented. In this section, we aim to draw inferences with respect to the results obtained.

To begin with fidelity, initially, a descriptive statistics methodology is employed to provide a detailed statistical analysis of the various elements the synthetic data that has been generated is to follow. The results obtained from each dataset are displayed and compared.

Type of Dataset	Parameter	response_address	resp_length	response_memory	resp_write_fun	response_memory_count	resp_read_fun	setpoint	control_mode	pump	result
Original data											
GAN data	Count	07010	07010	07010	97010	07010	07010	07010	97010	07010	07010
GMM data	Count	51015	31013	51015	31013	51015	51015	31013	31013	51015	31013
VAE data	1										
Original data		3.719	26.29	216.65	9.29	16.73	2.44	24.16	0.899	0.056	0.36
GAN data	Mana	1.14	48.77	66.48	2.88	5.15	1.98	39.77	0.97	0.953	0.053
GMM data	wean	3.714	26.29	216.4	9.286	16.71	2.34	24.26	0.899	0.033	0.37
VAE data	1	3.196	40.73	185.2	7.621	14.24	2.649	34.318	0.589	0.047	0.63
Original data		1.021	26.56	59.5	2.55	4.59	0.893	14.32	0.991	0.23	0.482
GAN data	84.3	0.512	12.12	27.07	1.19	2.11	0.193	8.101	0.17	0.211	0.224
GMM data	314	1.029	26.77	59.99	2.574	4.63	0.903	14.509	1.026	0.238	0.483
VAE data	1	0.652	14.38	32.32	1.382	2.565	0.490	9.645	0.539	0.211	0.2249

Table 4.11: Cumulative results of descriptive statistics

Using the Discriptive Statistics alone it is evident that the GMM model generated the synthetic data that is most alike to the original dataset. This test mainly handles the primary statistical features that the model is trained to replicate. The Synthetic Data Generation Model is expected to generate data that is capable of substituting the original data, thus the primary function of replicating or producing results close to that of the original data is paramount. Supplementing the results of the descriptive statistics methodology tests to provide additional information on the fidelity of the synthetic data are depicted in Tab.13.

Table 4.12: Cumulative results of Mahalanobis Distance and the	ne T2	test's p-value
--	-------	----------------

Model	Mahalanobis Distance Mean	Standard deviation	T ² test p-value
Original	71.6008	61.567	
GAN	43.598	28.625	0.012
GMM	72.606	62.353	1.11e-16
VAE	59.472	43.776	4.66e-11

The Mahalanobis Distance depicts the overall distance between two datasets, as per the tests definition the closer the mean distance to the original data the more significant the similarity. And to supplement the result of this test, a T2 test is also performed where the p-value is taken into consideration. The null hypothesis taken into consideration when performing the T2 test is that the two datasets are from the same distribution for p-value approaches 1. Thus using the scores of this test it can be seen that the GMM model possesses a p-value that is closest to 0, followed by the VAE model and then the GAN Model. This pattern is similarly replicated in the results of the Mahalanobis Distance measurements as well.

Next, the evaluation procedure that analyses Privacy and information Preservation is performed using the help of the MI score and the Wasserstein Distance. The MI Score in particular is useful when it comes to comparing the information content of the original dataset and the synthetic dataset. It helps to determine how well the synthetic data captures the information present in the original data. A higher MI score indicates a stronger relationship and greater similarity between the variables or datasets being compared. The Wasserstein Distance is often employed in synthetic data evaluation to assess the similarity between the probability distributions of the original and synthetic datasets. The MI score and the Wasserstein Distance show a similar performance trend when it comes to the synthetic datasets. In these tests, the results on Tab 4.12 are the absolute differences between the scores for the original and the synthetic dataset. The GMM model possesses the greatest information preservation capabilities and also displays the greatest similarity when it comes to the distribution of the dataset. This information preservation capability is further explored when considering the KDE plots that can determine how the synthetic data is distributed when compared with the original data. The results of the KDE test can be seen in Fig 4.1,4.2 and 4.3. The distribution characteristics shown by the GMM model are very similar to the original data, this could mean that the data has fulfilled one of the main purposes of generating synthetic data ie. to substitute the original data. The VAE model generates synthetic data that follows a similar pattern to that of the original model where it imitates the peaks and lulls but also provides a diverse take on the dataset, meaning that the dataset contains a variety of use case scenarios that would be present in a controlled environment such as the original dataset. On the other hand, the GAN models result is not as what is expected as it produces a diverse dataset that does not have a lot of similarity with the original dataset.

Continuing on with Diversity and Generalization, this is explored in depth with

Model	MI Score Difference	Wasserstein distance Difference
GAN	0.77	0.534
GMM	0.109	0.106
VAE	0.44	0.371

Table 4.13: Cumulative results of MI Score and Wasserstein Distance.

the assistance of Histogram plots, Scatter plots, and the KDE plot from the previous section Fig 4.1-3, Fig 4.5-7. The results obtained from these plots display a continuing pattern of GMM imitating the original dataset with the greatest likeliness, with the VAE model displaying variety and diversity but also maintaining a similar pattern, while the GAN models fall behind these models in this regard. To provide conclusive evidence of the above result, a Kolmogorov-Smirnov test is also performed. The Kolmogorov-Smirnov test can be used to determine whether the distribution of the synthetic dataset matches that of the original dataset. A smaller Kolmogorov-Smirnov statistic suggests a higher similarity between the distributions, indicating that the synthetic data is more representative of the original data. The results of this test can be seen in Tab 4.9.

The need for diversity is a growing need in the current day society where ICS datasets are of immense value and use. Real-time data lacks adverse case scenarios, where the ICS detects the signals present during the other end of the spectrum. This case scenario is least explored and not anticipated as much, therefore in accordance with the original data, the VAE Model would perform the best with regards to diversity while also having a good performance when it comes to predictive analysis. This can be seen in the previous subsection where interpretability and utility are discussed.

In this case, the experiments performed are by training the model on a predictive algorithm to determine its usability and performance. The models that have been trained on a decision tree are evaluated based on their predictive performance. Another version of the test is also performed to test the predictive capability of the model when trained on a dataset that possesses the combined result of the original dataset and the synthetic dataset of the respective model. The results proved conclusive as the GMM model performs the best with very high accuracy and performance, followed by the VAE model and GAN model respectively which could be seen in Fig.13. The second test also proved insightful as could be seen in Tab.11. The next test performed is to analyze the feature importance and if the weights assigned to each feature per model is similar or not. This can be done using a Python extension and the results can be seen in Fig.14. The feature importance values although not the same in nature, still do follow a similar pattern as above.

4.6 Discussions and Remarks

Different methods of generating high quality synthetic data for Industrial Control Systems (ICS) (GAN, VAE and the GMM models) was evaluated using visual analytics and statistical techniques. This research aimed on addressing a few factors such as the need for realistic and diverse datasets, datasets that have high levels of information preservation, and a suitable framework for evaluating the generated synthetic data. This thesis used a comprehensive comparative analysis to determine the best method.

An evaluation framework allowing us to quantify the similarity between synthetic and real datasets, measure data fidelity, evaluate information preservation, and assess the diversity and representativeness of the synthetic data was developed. We gained a deeper understanding of the characteristics of the synthetic data and made informed decisions regarding its suitability for different research and application contexts. The evaluation framework developed in this research facilitated the assessment of the synthetic data quality from various perspectives. The fields that have the highest need for synthetic data are evaluated based on their preference. Fidelity measures such as the Mahalanobis distance and the T2 test allowed for quantification of similarity between the original dataset and the synthetic dataset. Information preservation metrics such as the MI score and the KDE plots provided an accurate representation of the dataset's capability to preserve the essential information in the synthetic data. t-SNE plot visualizations aids in evaluating their interpretability and utility with machine learning techniques.

Through our comparative analysis, we evaluated three prominent techniques for generating synthetic data: Generative Adversarial Networks (GANs), Variational Autoencoders (VAEs), and Gaussian Mixture Model (GMM). Each technique was assessed in terms of its effectiveness, performance, and applicability to ICS applications. Our findings indicate that the Gaussian Mixture Model (GMM) outperformed the other techniques in terms of fidelity and performance. It demonstrated the highest ability to capture the statistical properties and patterns of real-world ICS datasets, making it the ideal choice for generating high-quality synthetic data in ICS.

One of the major research goals this thesis set out to achieve was to identify a synthetic data generation model that could aid researchers overcome the limitations of privacy and provide synthetic data that is identical to the original dataset. From the comparative analysis it can be seen that the GMM model excels in replicating and generating synthetic data that is extremely close to the original dataset over its competitors. The GMM model demonstrated a strong performance in capturing the statistical analysis properties of the original data, thus making it the ideal candidate in order to fulfil the primary study objective. The research also identified research gaps and challenges in the domain of generation and evaluation of synthetic data for ICS datasets and the need for further exploration of privacy and such information preservation techniques, capabilities in capturing dependencies and also addressing concerns regarding utility and scalability.

Information preservation is a key component of synthetic data generation. It is important for the synthetic data to preserve the characteristics as much possible from the original dataset. However, it is also important to have the capability to manipulate the synthetic data generation in such a way that it also involves improving adverse case scenarios to the dataset in question. This will enable to decrease the bias involved in prediction or analysis of synthetic data. In this case the VAE model excels in preserving the data and as well as generating a diverse dataset followed by the GAN models. Thus, this thesis provides valuable insights and aims to be a foundation to generate high-quality synthetic data for ICS datasets and provide an evaluation framework that is freely customizable to the user's everyday needs and also contributes to the advancements in these areas.

Chapter 5

Conclusion

In conclusion, through the comparative analysis of the three models the GAN, VAE, and the GMM models, it was observed that the CTGAN model performs well in enabling fine-grained control over the data generation. However, it is seen that the model requires further fine-tuning to be a suitable substitute for the original dataset or to create highly diverse datasets that could serve as an unbiased class of data. The outcomes of this thesis contribute to the advancement of the field of ICS research along with providing valuable insights into the strengths and weaknesses of the generation of synthetic data. Researchers can leverage this knowledge to select the right methods based on their needs and application processes. The availability of such high-quality datasets will assist people in future endeavors as high-quality sensor datasets will be readily available for research purposes and not be inhibited due to privacy concerns. This would also enable the development of algorithms tailored for ICS datasets and also improve the system designs and enhanced security mechanisms of the ICS.

5.1 Future Work

It is worth noting that this research is not without limitations. While our evaluation framework provided a comprehensive analysis, there may be additional factors or metrics that could be considered in future studies. Furthermore, the choice of synthetic data generation technique may vary depending on the specific characteristics of the ICS dataset and the research objectives.

It is also recommended to work on other forms of synthetic datasets as these SCADA-based networks are popular in not only GAS pipeline datasets but also in water and natural gas as well. This indicates that the evaluation framework can be applied to a wide variety of datasets to check its integrity and usefulness. It is also imperative to explore Transformer based models, diffusion models and other such hybrid models to diversify the evaluation result. Continual evaluation of the ICS datasets and improvement of the framework is paramount for success to address the future challenges experienced in the various synthetic data generation models.

Thus, this thesis provides valuable insights and aims to be a solid foundation to generate high-quality synthetic data for ICS datasets and provide an evaluation framework that is freely customizable to the user's everyday needs and also contributes to the advancements in these areas.

Bibliography

- A. Goncalves et al., "Generation and evaluation of synthetic patient data," BMC Medical Research Methodology, vol. 20, no. 1, pp. 1–40, 2020.
- [2] D.Upadhyay and S.Sampalli, "Scada (supervisory control and data acquisition) systems: Vulnerability assessment and security recommendations," *Computers* and Security, vol. 89, p. 101666, 2020.
- [3] T. Akhtar and B. Gupta, "Analysing smart power grid against different cyber attacks on scada system," *International Journal of Innovative Computing and Applications*, vol. 12, no. 4, pp. 195–205, 2021.
- [4] M. Hittmeir, A. Ekelhart, and R. Mayer, "On the utility of synthetic data: An empirical evaluation on machine learning tasks," in *Proceedings of the 14th International Conference on Availability, Reliability, and Security*, 2019.
- [5] K. E.Emam, "Seven ways to evaluate the utility of synthetic data," *IEEE Secu*rity & Privacy, vol. 18, no. 4, pp. 56–59, 2020.
- [6] D. L. Banks, P. D. Ruggiero, and A. B. Ruggiero, "Synthetic data in computational statistics," in *Handbook of Computational Statistics*, pp. 931–951, Springer, 2004.
- [7] M. Zhang, Z. Ren, Y. Xiang, and S. Mao, "A statistical framework for evaluating synthetic datasets for industrial control systems," *IEEE Transactions on Industrial Informatics*, vol. 16, no. 7, pp. 4450–4461, 2020.
- [8] A. Alshehri, R. Mahapatra, and K. McLaughlin, "Synthetic ics sensor data generation using gans and lstm networks," *IEEE Transactions on Industrial Informatics*, vol. 16, no. 4, pp. 2563–2573, 2020.
- [9] J. S. Breese and D. Heckerman, "Decision-theoretic debugging," in *Proceedings* of the 13th conference on Uncertainty in Artificial Intelligence, pp. 55–64, 1997.
- [10] K. Zhang, Y. Wu, and X. Hu, "Synthetic ics datasets generation framework for training intrusion detection systems," *IEEE Transactions on Industrial Informatics*, vol. 15, no. 11, pp. 6013–6022, 2019.
- [11] S. Song et al., "Statistical evaluation of synthetic power system data," IEEE Transactions on Power Systems, vol. 33, no. 3, pp. 2817–2825, 2022.
- [12] S. Yasakethu and J. Jiang, "Intrusion detection via machine learning for scada system protection," in 1st International Symposium for ICS & SCADA Cyber Security Research 2013 (ICS-CSR 2013) 1, pp. 101–105, 2013.

- [13] L. Chen, Y. Li, X. Deng, Z. Liu, M. Lv, and H. Zhang, "Dual auto-encoder gan-based anomaly detection for industrial control system," *Applied Sciences*, vol. 12, no. 10, p. 4986, 2022.
- [14] L. Xu and K. Veeramachaneni, "Synthesizing tabular data using generative adversarial networks," arXiv preprint arXiv:1811.11264, 2018.
- [15] P. Chen, Z. Yu, and Y. Xiao, "Synthetic data generation for industrial control systems using bayesian networks," *IEEE Transactions on Industrial Informatics*, vol. 13, no. 6, pp. 3124–3134, 2017.
- [16] J. Wang, X. Liu, and Y. Liu, "Synthetic data generation for industrial control systems using variational autoencoders," *Sensors*, vol. 20, no. 21, p. 6247, 2020.
- [17] Y. Wu, C. Xu, K. Zhang, and X. Hu, "Visualization techniques for evaluating synthetic industrial control system datasets," *IEEE Transactions on Visualization and Computer Graphics*, vol. 25, no. 1, pp. 949–958, 2019.
- [18] H. Zou, L. Huang, and H. H. Ali, "Generative adversarial networks-based synthetic data generation for industrial control systems," *IEEE Access*, vol. 7, pp. 45142–45152, 2019.
- [19] L. Xu, M. Skoularidou, A. Cuesta-Infante, and K. Veeramachaneni, "Modeling tabular data using conditional gan," Advances in Neural Information Processing Systems, vol. 32, 2019.
- [20] E. Hodo et al., "Metrics for evaluating the quality of synthetic intrusion detection system datasets," Computers & Security, vol. 68, pp. 18–31, 2017.
- [21] J. Chen, C. Yang, X. Huang, C. Xiao, and G. Yang, "Differentially private synthetic data generation for industrial control systems," *IEEE Transactions on Industrial Informatics*, vol. 17, no. 3, pp. 2019–2029, 2021.
- [22] J. Ho, A. Jain, and P. Abbeel, "Denoising diffusion probabilistic models," in Advances in Neural Information Processing Systems (H. Larochelle, M. Ranzato, R. Hadsell, M. Balcan, and H. Lin, eds.), vol. 33, pp. 6840–6851, Curran Associates, Inc., 2020.
- [23] Y. Li, Z. Tang, X. Zhu, and F. Wu, "Generating synthetic datasets for anomaly detection in industrial control systems," in *Proceedings of the 15th IEEE International Conference on Networking, Sensing, and Control*, pp. 1–6, 2018.
- [24] C. T. Symons and J. M. Beaver, "Nonparametric semi-supervised learning for network intrusion detection: combining performance improvements with realistic in-situ training," in *Proceedings of the 5th ACM workshop on Security and artificial intelligence*, pp. 49–58, 2012.

- [25] B. Zhang, X. Liu, and J. Wang, "Synthetic data generation for industrial control systems using generative adversarial networks," *IEEE Access*, vol. 8, pp. 12370– 12380, 2020.
- [26] L. Qian, X. Zhu, Y. Li, and F. Wu, "Synthetic data generation for anomaly detection in industrial control systems using generative adversarial networks," *IEEE Transactions on Industrial Informatics*, vol. 15, no. 1, pp. 487–495, 2018.
- [27] X. Liu *et al.*, "Evaluating the utility of synthetic data for differentially private time series classification," *IEEE Transactions on Knowledge and Data Engineering*, vol. 32, no. 8, pp. 1503–1516, 2020.
- [28] A. Salem et al., "Evaluating the privacy and utility of synthetic social network data," ACM Transactions on Privacy and Security, vol. 21, no. 2, pp. 1–28, 2018.
- [29] W. K. Wong et al., "Visual data comparison and its application in evaluating synthetic data," *IEEE Transactions on Visualization and Computer Graphics*, vol. 22, no. 1, pp. 369–378, 2016.
- [30] Y. Zhang *et al.*, "Evaluating the utility of generative adversarial networks in synthetic medical data generation," *Artificial Intelligence in Medicine*, vol. 82, pp. 45–56, 2016.
- [31] E. Monteiro et al., "A visual analytics approach for synthetic ics data evaluation," *IEEE Transactions on Visualization and Computer Graphics*, vol. 26, no. 1, pp. 979–988, 2020.
- [32] M. Noorizadeh, M. Shakerpour, N. Meskin, D. Unal, and K. Khorasani, "A cybersecurity methodology for a cyber-physical industrial control system testbed," *IEEE Access*, vol. 9, pp. 16239–16253, 2021.
- [33] J. Giraldo et al., "Evaluation of synthetic power system data considering the system behavior," *IEEE Transactions on Smart Grid*, vol. 10, no. 6, pp. 6267– 6276, 2019.
- [34] J. Liu, H. Gao, H. Li, Y. Song, and W. Han, "Synthetic data generation using deep learning for anomaly detection in industrial control systems," *Applied Sciences*, vol. 10, no. 10, p. 3429, 2020.
- [35] A. E. Howe, H. Fei, J. S. Pupillo, and A. J. Medford, "Synthesized measurement data in support of nuclear safeguards decision making," in *International Topical Meeting on Nuclear Safeguards and Security*, 2013.
- [36] D. Elkan, "The foundations of cost-sensitive learning," in International Joint Conference on Artificial Intelligence, pp. 973–978, 2001.
- [37] K. S. Deepak, M. Chakraborty, and R. Jana, "Cyber-physical system security: A case study on the stuxnet worm," in 2011 Annual IEEE India Conference, pp. 1–6, IEEE, 2011.

[38] L. Xu, M. Skoularidou, A. Cuesta-Infante, and K. Veeramachaneni, "Modeling tabular data using conditional gan," Advances in Neural Information Processing Systems, vol. 32, 2019.